Author's response to reviews

Title: Granulomatous hepatitis due to Bartonella henselae Infection in an immunocompetent patient.

Authors:

Thomas R VanderHeyden Jr. (tvh25@yahoo.com)
Claus J Fimmel (cfimmel@lumc.edu)
Sherri L Yong (syong@lumc.edu)
Edward B Breitschwerdt (ed_breitschwerdt@ncsu.edu)
Jorge P Parada (jparada@lumc.edu)
Amanda R Mihalik (amihalik@lumc.edu)

Version: 5 Date: 29 November 2011

Author's response to reviews:

Enclosed you will find the most recent edition to the manuscript A case of granulomatous hepatitis due to Bartonella henselae in an immunocompetent patient. In the revised submission, you will note there has been an addition of content as PCR testing was performed on the original surgical sample. This is summarized in Figure 5 (hepatic mass). The manuscript has also undergone changes to make the background more concise with an expanded conclusion. We thank you for extended timeline to complete this revised manuscript. We believe you will find these changes appropriate and also interesting. We look forward to hearing from you.